The disease is relatively rare. Although the number of medical records is not much, the author's thinking is clear and has clinical significance.

We thank the reviewer for the positive comments on our manuscript. It means that we have found an interesting topic that was processed correctly and accurately. Thank you once again.

This a good manuscript that examines the difficult issue of CDH during pregnancy. The aim is clearly stated from the title on. The methodology is correct. The results are well analyzed and displayed. The discussion is too long but correct and the references are good. I suggest to abridge a little bit the discussion, but otherwise I would recommend acceptance.

We also thank reviewer#2 for a similar view of the manuscript as reviewer#1. We shortened the discussion for a minor part. The further reduction would lead to unreadable and hardly understandable text.

All changes are highlighted, and comments are put in a Word document. A CONCLUSION section was added.